Daily Status Sat 20-Oct-12

20-Oct-12

Weather & Ops

Clear and pretty calm in the am. Wspd at .5m was <2 and 1m ~2.5 with
the coolest site being 11 in the bottom of the confluence. However
temps have been staying up around 10 + 4ish so maybe still too warm
for preferred event.

Communications are all up and files are growing. Everyone is reachable.
Gully is humming along, disks show /media/isfsO 75% and pocketec was at
40% but swapped out to bring hr data back to Boulder. New pocketec at 1%

Sensors & Issues

Thank goodness for the qc table and plots. Everything appears up.
Very few diags on sonics. Krypton voltages still up. Battery
voltages good, and they should be after all the sun the last few days.

To Dos

Bring Flux computer back on line

Crew change: jm leaving this morning, SteveS/ChrisG arriving tomorrow

morning.

Separate list also on wiki

Mini-List for ChrisG: shoot another set of boom angles, finish

protective greasing trh wagon-wheels for al4-al8. Put high-q

conductive grease on all din connector pins. Keep eye on battery

charging but it seems ok. Clean rad, licors and eventually kh20. Do

we want a soil sample? One TRH-i2c cable and pc-console cable in base trailer to repair.
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